Author's response to reviews

Title: The impact of a standardised intramuscular sedation protocol for acute behavioural disturbance in the emergency department

Authors:

Leonie A Calver (leonie.calver@mater.health.nsw.gov.au)
Michael A Downes (Michael.Downes@mater.health.nsw.gov.au)
Colin B Page (cpage@bigpond.net.au)
Jenni L Bryant (jenni.bryant@mater.health.nsw.gov.au)
Geoffrey K Isbister (geoff.isbister@gmail.com)

Version: 6 Date: 23 June 2010

Author's response to reviews:

Changes made.
Except, there are no references in the abstracts, the numbers 1, 4 and 5 in parentheses refer to the number of cases with each of adverse effects - NB> this is the same in the main result section so please don't make these references either.